Author's response to reviews

Title: Efficacy of a web- and text messaging-based intervention to reduce problem drinking in young people: Study protocol of a cluster-randomised controlled trial

Authors:

Severin Haug (severin.haug@isgf.uzh.ch)
Tobias Kowatsch (tobias.kowatsch@unisq.ch)
Raquel Paz Castro (raquel.paz@isgf.uzh.ch)
Andreas Filler (afiller@ethz.ch)
Michael P Schaub (michael.schaub@isgf.uzh.ch)

Version: 2
Date: 28 July 2014

Author's response to reviews:

Cover Letter with additional information addressing editor requests from 15/07/2014.